\begin{thebibliography}{10}

\bibitem{adar00free}
Eytan Adar and Bernardo~A. Huberman.
\newblock Free riding on gnutella.
\newblock {\em First Monday}, September 2000.

\bibitem{BAJ99}
Albert-La\'azl\'o, R\'eka Albert, and Hawoong Jeong.
\newblock Scale-free characteristics of random netwoks: The topology of the
  world wide web.
\newblock {\em submitted to {E}lsevier {Preprint}}, August 1999.

\bibitem{beth94valuation}
T.~Beth, M.~Borcherding, and B.~Klein.
\newblock Valuation of trust in open networks.
\newblock {\em Lecture Notes in Computer Science}, 875:3--??, 1994.

\bibitem{Bla96}
Matt Blaze.
\newblock Oblivious key escrow.
\newblock In {\em Proc. Workshop on Information Hiding}, number 1174 in LNCS,
  pages 334--343. Springer-Verlag, 1996.

\bibitem{BFL96}
Matt Blaze, Joan Feigenbaum, and Jack Lacy.
\newblock Decentralized trust management.
\newblock In {\em Proc. 17th Symposium on Security and Privacy}, pages
  164--173, Los Alamitos, 1996. IEEE Computer Society Press.

\bibitem{Bra97}
Marc Branchaud.
\newblock A survey of public key infrastructures.
\newblock Master's thesis, McGill University, Dept. of Computer Science, March
  1997.

\bibitem{CL99}
Miguel Castro and Barbara Liskov.
\newblock Practical byzantine fault tolerance.
\newblock In {\em Proc. Third Symp. on Operating Systems Design and
  Implementation}, New Orleans, February 1999.

\bibitem{chaum89untraceable}
D.~Chaum, A.~Fiat, and M.~Naor.
\newblock Untraceable electronic cash (extended abstract), 1989.

\bibitem{crary99typed}
Karl Crary, David Walker, and Greg Morrisett.
\newblock Typed memory management in a calculus of capabilities.
\newblock In {\em Conference Record of {POPL} 99: The 26th {ACM}
  {SIGPLAN}-{SIGACT} Symposium on Principles of Programming Languages, San
  Antonio, Texas}, pages 262--275, New York, NY, 1999.

\bibitem{chord:hotos}
Frank Dabek, Emma Brunskill, M.~Frans Kaashoek, David Karger, Robert Morris,
  Ion Stoica, and Hari Balakrishnan.
\newblock Building peer-to-peer systems with {C}hord, a distributed lookup
  service.
\newblock In {\em Proceedings of the 8th {W}orkshop on {H}ot {T}opics in
  {O}perating {S}ystems ({HotOS-VIII})}, Schloss Elmau, Germany, May 2001.
  {IEEE} {C}omputer {S}ociety.

\bibitem{dingledine00free}
Roger Dingledine, Michael~J. Freedman, and David Molnar.
\newblock The free haven project: Distributed anonymous storage service.
\newblock In {\em Workshop on Design Issues in Anonymity and Unobservability},
  number 2009 in LNCS, pages 67--95, 2000.

\bibitem{dingledine-cascrep}
Roger Dingledine and Paul Syverson.
\newblock Reliable {MIX} cascade networks through reputation.
\newblock In {\em Proc. Financial Cryptography}, March 2002.

\bibitem{douceur-sybil}
John~R. Douceur.
\newblock The sybil attack.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\bibitem{dwork-spam}
Cynthia Dwork and Moni Naor.
\newblock Pricing via processing or combating junk mail.
\newblock In Ernest~F. Brickell, editor, {\em Advances in Cryptology -- CRYPTO
  '92}, number 740 in LNCS, pages 139--147. Springer-Verlag, 1992.

\bibitem{flake02self-organization}
Gary~William Flake, Steve Lawrence, C.~Lee Giles, and Frans Coetzee.
\newblock Self-organization of the web and identification of communities.
\newblock {\em IEEE Computer}, 35(3):66--71, 2002.

\bibitem{gay01language}
David Gay and Alexander Aiken.
\newblock Language support for regions.
\newblock In {\em {SIGPLAN} Conference on Programming Language Design and
  Implementation}, pages 70--80, 2001.

\bibitem{kleinberg-smallworld}
Jon Kleinberg.
\newblock The small-world phenomenon: An algorithmic perspective.
\newblock In {\em Proc. 32nd ACM Symp. on Theory of Computing}, 2000.

\bibitem{kleinberg99clever}
Jon~M. Kleinberg.
\newblock Authoritative sources in a hyperlinked environment.
\newblock {\em Journal of the {ACM}}, 46(5):604--632, 1999.

\bibitem{LSP82}
Leslie Lamport, Robert Shostak, and Marshall Pease.
\newblock The byzantine generals problem.
\newblock {\em {ACM} Transactions on Programming Languages and Systems},
  4(3):382--401, July 1982.

\bibitem{LA98}
Raph Levien and Alexander Aiken.
\newblock Attack resistant trust metrics for public key certification.
\newblock In {\em 7th {USENIX} Security Symposium}, San Antonio, Texas, January
  1998.

\bibitem{Mau96}
Ueli Maurer.
\newblock Modelling a public-key infrastructure.
\newblock In E.~Bertino, H.~Kurth, G.~Martella, and E.~Montolivo, editors, {\em
  Computer Security -- {ESORICS} '96}, number 1146 in LNCS. Springer Verlag,
  1996.

\bibitem{maymounkov-kademlia}
Petar Maymounkov and David Mazi\`eres.
\newblock Kademlia: A peer-to-peer information system based on the {XOR}
  metric.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\bibitem{agorics-incentive}
Mark~S. Miller and K.~Eric Drexler.
\newblock Incentive engineering: for computational resource management.
\newblock In Bernardo Huberman, editor, {\em The Ecology of Computation}, pages
  231--266. Elsevier Science Publishers/North-Holland, 1988.

\bibitem{agorics-markets}
Mark~S. Miller and K.~Eric Drexler.
\newblock Markets and computation: Agoric open systems.
\newblock In Bernardo Huberman, editor, {\em The Ecology of Computation}, pages
  133--176. Elsevier Science Publishers/North-Holland, 1988.

\bibitem{moreton-trust}
Tim Moreton and Andrew Twigg.
\newblock Trading in trust, tokens and stamps.
\newblock In {\em Proc. Workshop on Economics of Peer-to-Peer Systems}, June
  2003.

\bibitem{P2P}
Andy Oram, editor.
\newblock {\em Peer to Peer}.
\newblock O'Reilly \& Associates, 2001.

\bibitem{page98pagerank}
Lawrence Page, Sergey Brin, Rajeev Motwani, and Terry Winograd.
\newblock The {P}age{R}ank citation ranking: Bringing order to the web.
\newblock Technical report, Stanford University, 1998.

\bibitem{Pos99}
David Post, 1999.

\bibitem{operatingthetan}
ptsc (pseudonym).
\newblock The {C}hurch of {S}cientology's supremacy over the search term
  ``{S}cientology'' on {G}oogle, February 2002.

\bibitem{RS97a}
Michael Reiter and Stuart Stubblebine.
\newblock Path independence for authentication in large-scale systems.
\newblock In {\em Proceedings of the 4th {ACM} Conference on Computer and
  Communications Security}, 1997.

\bibitem{RS97b}
Michael Reiter and Stuart Stubblebine.
\newblock Toward acceptable metrics of authentication.
\newblock In {\em Proceedings of the 1997 {IEEE} Symposium on Security and
  Privacy}, 1997.

\bibitem{Ross67}
D.~T. Ross.
\newblock The {AED} free storage package.
\newblock {\em Communications of the ACM}, 10(8):481--492, August 1967.

\bibitem{chord:sigcomm01}
Ion Stoica, Robert Morris, David Karger, M.~Frans Kaashoek, and Hari
  Balakrishnan.
\newblock Chord: A scalable peer-to-peer lookup service for internet
  applications.
\newblock In {\em Proceedings of the {ACM} {SIGCOMM} '01 Conference}, San
  Diego, California, August 2001.

\bibitem{tarah92associating}
Anas Tarah and Christian Huitema.
\newblock Associating metrics to certification paths.
\newblock In {\em European Symposium on Research in Computer Security
  ({ESORICS})}, pages 175--192, 1992.

\bibitem{tofte97regionbased}
Mads Tofte and Jean-Pierre Talpin.
\newblock Region-based memory management.
\newblock {\em Information and Computation}, 1997.

\bibitem{zooko-mn}
Bryce Wilcox-O'Hearn.
\newblock Experiences deploying a large-scale emergent network.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\end{thebibliography}

\chapter{An attack-resistant name service}
\label{nameservice}

This chapter presents the design of an attack-resistant name service,
using the group trust metric described in Chapter~\ref{groupmetric} to
provide high assurance even under attack.

For the purpose of this chapter, name service is the problem of
finding the appropriate public key corresponding to a given name. This
problem is quite thorny in practice. Determining who owns the rights
to a given name is not well-defined. Rather, it is the subject of
social, economic, legal, and political issues. Real world CA's have
misbehaved. Cite VeriSign/Microsoft.

We address these issues by separating mechanism from policy. This
requires policy to be formally specified. Thus, the policy language
must be rich and flexible enough to accommodate the policies desired.

Probably the only workable policy is ``first-come, first-served,'' as
it's the only one that's understandable.

Choice of namespaces: flat, hierarchical, hash-based. Also cite SDSI
for relative namespaces.

\section{Related work}

There are many designs and implementations of systems to bind keys to
names. The most popular is the X.509 family of standards, loosely
synonymouse with the Public Key Infrastructure (PKI). In this
framework, authority for the namespace is assigned to one or more
\emph{certification authorities} (CA's), generally composed of
\emph{root CA's} that delegate authority over some or all of the
namespace to various subsidiary CA's. A consequence of this design
decision is the \emph{root vulnerability}---the compromise of any of
the root CA's leads to catastrophic security failure.

Root vulnerability is a serious concern. Even though individual CA's
may be well managed (with private keys reasonably well protected),
modern applications depending on PKI such as Web browsers ship with
dozens of root CA keys enabled. Further, the grouwing popularity of CA
software for off-the-shelf platforms (such as Entrust/PKI and
Entegrity's Notary) implies that CA's will be deployed in contexts
where it is difficult or impossible to protect the CA with a high
degree of assurance.

Often, the CA given root authority is implementing a fairly simple and
well defined policy. For example, VeriSign's Class 1 certificates are
issued on a first-come, first-served basis. Domain name registrars
issue second level domains on an effectively first-come, first-served
basis, with the possibility of revoking the name (and issuing it to
someone else) for non-payment of the bill, or as the result of a
dispute. Actual implementation of these policies is at the whim of
whoever holds the CA's private key. An attacker who succeeds in
compromising the CA key can reassign names completely at will, with no
regard to any policy.

It is possible to implement well defined policies and avoid root
compromise. The naming service presented in this chapter factors the
name service problem into to sub-problems: a \emph{policy language}
for expressing policies formally, and a \emph{distributed trusted
third party} for implementing the policies. In cases where the complex
blend of factors controlling ownership of names can be distilled into
a formal policy, our naming service can provide much higher assurance
with lower certification cost than existing PKI's.

A design factored in this way can only be successful if two goals are
met: the policy language should be rich enough to express a range of
policies useful in the real world, and the distributed trusted third
party should be trustworthy. Section~\ref{policylang} presents such a
policy language, and Section~\ref{dttp} presents a distributed network
based on an attack-resistant trust metric.

The policy language introduces the concept of \emph{asymmetry} between
initial registration of a name and updates to that name. In
traditional PKI designs, the authority granted to the CA is total, so
the CA has equal power to register and update name/key
bindings. However, there are many interesting asymmetrical policies
where the power to update is more restricted than the power to
register. The first-come, first-served policy is an extreme
example. Authority to register new names is trivially granted, but
authority to update existing names is never granted.

\section{Policy language}
\label{policylang}

The goals of the policy language are:

\begin{itemize}

\item Flexibility: different policies can be specified for different
parts of the namespace.

\item Low certification cost if security against unauthorized
registrations is not important.

\item High security if higher certification cost is tolerated.

\item Good security against unauthorized updates.

\item Low vulnerability to root compromise.

\end{itemize}

Meeting all these goals requires a moderately complex language. In
practice, a simpler policy, such as first-come first-served, may be a
better choice. First-come first-served is poor at assigning names
consistent with some other concept of ownership (such as trademarks),
but its limitations are easily understood and analyzed.

Before defining the policy language itself, we define the space of
\emph{names} and \emph{requests} over which the policies are
defined. Names are similar to familiar Internet hostnames and e-mail
addresses, such as ``raph@cs.berkeley.edu''. As in DNS, names are
hierarchically structured. Thus, the parent of
``raph@cs.berkeley.edu'' is ``cs.berkeley.edu'', its parent in turn is
``berkeley.edu'', and so on. Names with children are \emph{domains.}

Each request is either a \emph{registration} or an \emph{update}
request, and in either case contains a name, a value (such as a public
key) to be bound to that name, and policies to be associated with that
name. In addition to standard names, there are special namespaces for
policy fragments and groups. Each request is also accompanied by a
signature from zero or more public keys.

This design does not address read access to the database. Always
allowing lookups is reasonable and consistent with existing Internet
practice. Read access control is a plausible extension, however.

A request is processed as follows:

\begin{itemize}

\item The signatures on the request are verified, resulting in a list
of public keys that signed the request.

\item The body of the request is used to retrieve an appropriate policy
from the database.

\item The policy is evaluated over the list of public keys signing the
request, resulting in a boolean value.

\item If the boolean value is \emph{true,} then the database is
updated as per the request.

\end{itemize}

This workflow is similar to that of PolicyMaker\emph{BFL96} and other
trust management systems.

For registration requests, the appropriate policy is the
\emph{registration policy} of the parent. For update requests, the
appropriate policy is the \emph{update policy} of the name. The update
policy for a name is fixed upon registration of the name. It cannot
itself be directly updated. The registration policy for a domain can
be updated.

The policy language is defined as follows:

\[
\begin{array}{rcl}
policy & ::= & int\ 'of'\ group \\
       & | &  policy\ '\wedge'\ policy \\
       & | & policy\ '\vee'\ policy \\
       & | & 0 \\
       & | & 1 \\
       & | & '*'\ policy\mbox{-}name
\end{array}
\]


\[
\begin{array}{rcl}
group & ::= & '\{'\ key\ '\}' \\
        & | & group\ '\cup'\ group \\
        & | & group\ '\cap'\ group \\
        & | & '*'\ group\mbox{-}name
\end{array}
\]

\[
\begin{array}{rcl}
key & ::= & key\mbox{-}hash \\
      & | & '*'\ key\mbox{-}name
\end{array}
\]

\[
\begin{array}{rcl}
policy\mbox{-}name & ::= & id\ '?'\ name
\end{array}
\]

\[
\begin{array}{rcl}
group\mbox{-}name & ::= & id\ '@@'\ name
\end{array}
\]

\[
\begin{array}{rcl}
key\mbox{-}name ::= name
\end{array}
\]

As stated above, the goal of this policy constraint language is to
enable a DNS-like naming hierarchy with a flexible, varying degree of
control. A particular goal is to preserve the security of delegated
sections of namespace even when the nodes in control of the root, or
high-level domains, are compromised. At the same time, if a delegated
name turns out to be improperly granted, it should be possible to
recall it.

The policy constraint language provides a mechanism to balance these
two goals. It easy expresses both the first come, first served policy,
by trivially granting registration requests, but never granting update
requests. Similarly, the policy of central control is easily
represented by symmetrical registration and update policies.

Evaluation of whether a given policy meets a policy constraint
expressed in the policy constraint language is a co-NP complete
problem. Here we present a brief proof outline in the form of a
reduction to 3-SAT, which is a well known NP-complete problem.

Present pseudocode algorithm for implementing policy constraint
language.

\section{Implementation of naming service}

In order to avoid a single point of failure, we choose a peer-to-peer
network architecture for the implementation of the naming service. For
simplicitly, we choose a two-level architecture, with a relatively
small number of ``server'' nodes, and most clients contacting the
server nodes to retrieve names and certificates.

All servers have knowledge of all other servers. Thus, notification of
server joins and leaves are broadcast operations. When the number of
servers is on the order of the square root of the total number of
peers, network traffic is optimized. This square-root behavior is
neither as bad as flat broadcast networks, such as the original
Gnutella protocol, nor as good as the logarithmic performance of
systems such as Chord\cite{chord:sigcomm01}.

Not all nodes store records for all names registered in the
system. Rather, there is a ``responsible server'' relation, generally
based on a hash function so that the set of responsible servers for a
given name is effectively random. An average of 10 to 20 servers
should be responsible for each name. This average is a tunable
parameter. As it increases, security improves, but overall network
traffic also increases.

There are essentially two different authorization decisions in the
system. The first, evaluated by nodes when processing registration and
update requests, is whether the request is valid. The second,
evaluated by clients when performing queries, is whether the node
responding to the query is trustworthy to provide the correct answer.
Our design mirrors the fundamental asymmetry of these two decisions.
Essentially, for \emph{mutation} requests, clients send the request to
all responsible nodes for the name, regardless of whether such nodes
are to be considered trustworthy. However, for \emph{query} requests,
clients only consider responses from nodes which are both responsible
for the name and considered trustworthy.

Our design mixes up this clear separation a bit, because nodes
processing mutation requests also function in the client role when
they retrieve the policy and policy constraint associated with the
parent name.

\subsection{Responsible servers}

This subsection presents a simple, effective relation for determining
which nodes are responsible for which names.

Each server has an ID, which is the 160-bit SHA-1 hash of its public
key. Similarly, the SHA-1 hash of any name gives a value in this
space, ie the id of name $x$ is $H(x)$.

We define a ``distance metric'' $d(n_1, n_2)$ to be the absolute value
of $n_1 - n_2$ interpreted using 160-bit two's complement arithemetic.
This metric induces a ``ring'' topology on the space of ID's.

With this definition, a server with id $n_s$ is responsible for name
$x$ iff $d(n_s, H(x)) < \Delta d$, where $\Delta d$ is a tunable
parameter of the system. A reasonable target is that 10 to 20 good
servers are responsible for any given name, i.e. $\Delta d$ is equal
to 5 to 10 times $2^{160}$ divided by the number of good servers.

\subsection{Mutation requests}

In this subsection, we describe in detail the algorithm used by nodes
to authenticate mutation requests. Overall, this authentication process
strongly resembles PolicyMaker\cite{BFL96}.

There is no consistency guarantee for mutation requests. In
particular, two registrations coming in at roughly the same time may
lead to inconsistency, with different nodes giving different answers
to the same query. Improving consistency is an interesting research
direction. Perhaps the most profitable approach is to adapt the
Byzantine fault tolerance research\cite{CL99}.

\subsection{Query requests}

In this subsection, we describe and analyze the algorithm used by
clients to validate responses to query requests. This process is
strongly dependent on a group trust metric, and we discuss how the
security properties of the trust metric affect the security of the
system as a whole.

The client's first step is to determine a set of responsible and
trustworthy servers. In our design, we simply determine the set of all
trustworthy servers first, then apply the ``responsibility'' relation
as a filter. It is tempting to consider only a subset of all
trustworthy nodes. Such a design might be considerably more efficient
(as is demonstrated by the Distributed Hash Tree work such as
Chord\cite{chord:hotos} and Kademlia\cite{maymounkov-kademlia}), but
unfortunately the security of the group trust metric as described in
Chapter~\ref{groupmetric} depends on knowledge of the entire trust graph.

Armed with this list (which is expected to contain on the order of 10
or 20 nodes), the client sends the request separately to each node.

Finally, as the responses come trickling back, the client waits until
a majority of nodes queried provide a consistent response. If no single
response is in the majority, then the lookup is considered to have
failed, with no trustworthy binding to that name.

Note that in the common case, when all nodes provide the same answer,
the client accepts the binding as soon as the earliest half of nodes
respond. Thus, individual server failures are tolerated, and request
latency is equal to the median latency of server nodes.

\subsection{Security analysis}

The voting semantics of the validation of responses to queries is a
good match to the guarantees provided by the group trust metric.

Assuming no conflicts during the registration phase, then all good
nodes responsible for a name will yield the same (``correct'') answer
to a query. Thus, if at least one half of the subset of responsible
nodes accepted by the trust metric are good, then the correct answer
will be obtained.

The group trust metric can guarantee that a fraction of the total
nodes accepted is good, but this is not the same as guaranteeing that
a fraction of the accepted nodes responsible for a name is
good. Indeed, it is likely that an attacker would be able to mount a
successful attack against a single name at low cost.
